Author’s response to reviews

Title: Effects of self-monitoring of glucose in non-insulin treated patients with type 2 diabetes: design of the IN CONTROL-trial

Authors:

Uriëll L Malanda (u.malanda@vumc.nl)
Sandra DM Bot (s.bot@vumc.nl)
Piet J Kostense (pi.kostense@vumc.nl)
Frank J Snoek (fj.snoek@vumc.nl)
Jacqueline M Dekker (jm.dekker@vumc.nl)
Giel Nijpels (g.nijpels@vumc.nl)

Version: 5 Date: 23 April 2009

Author’s response to reviews:

April 23, 2009

Dear Editor,

In this letter I present an overview of the final corrections made.

Final corrections:
# In the ‘Abstract’ under heading ‘Methods/design’ randomised is corrected in randomized
# The paragraphs ‘Process evaluation and quality assurance’ and ‘Sample size’ have been grammatically corrected.

I hope the revised manuscript addresses your requirements.

With kind regards,

Uriëll L. Malanda
Corresponding author